\documentstyle{article}
\begin{document}
\cite{chicago,texbook,prime-number-theorem,1234567890)(@!`['";:]|<>./?-=+_$^&}
\cite{others,accented}
\bibliographystyle{plain}
\bibliography{cite}
\end{document}
